Author's Response To Reviewer Comments

Clo<u>s</u>e

We greatly appreciate the opportunity given to review our manuscript. We hope we have been able to address the editor's concerns and that our manuscript is now suitable for publication in GigaScience.

Clo<u>s</u>e